Author's response to reviews

Title: Individual and Community Level Determinants of Childhood Full Immunization in Ethiopia: A Multilevel Analysis

Authors:

Samir A Abadura (samirawol2007@gmail.com)
Wondwosen T Lerebo (kidwonyt4@gmail.com)
Usha Kulkarni (ushakulkarni@hotmail.com)
Zeleke A Mekonnen (zelekeabebaw@yahoo.com)

Version: 9 Date: 16 September 2015

Author's response to reviews:

Comments: *Author’s Contribution: Please correctly format your author's contribution section.

Reply: We have clearly rectified based on the format in the main manuscript on line 446-51 page 18-9: Authors’ Contribution

SA conceived of the study, participated in its design and coordination, drafted the manuscript, initiated the research, carried out the statistical analysis, interpreted the results and drafted the final manuscript. WL, UK and ZM participated in the design and coordination of the study, revised the proposal and guided the statistical analysis and write up of the manuscript. All authors read and approved the final manuscript.